Author’s response to reviews

Title: High body mass index is not associated with atopy in schoolchildren living in rural and urban areas of Ghana

Authors:

Irene A Larbi (ialarbi@gmail.com)
Kerstin Klipstein-Grobusch (kerstin.klipstein-grobusch@wits.ac.za)
Abena S Amoah (aamoah@noguchi.mimcom.org)
Benedicta B Obeng (baffoah@yahoo.com)
Michael D Wilson (wilsonmi@who.int)
Maria Yazdanbakhsh (m.yazdanbakhsh@lumc.nl)
Daniel A Boakye (dboakye@noguchi.mimcom.org)

Version: 5 Date: 11 June 2011

Author’s response to reviews:

Dear Editor,

In response to the editorial comments the title for figure 2 was amended (Figure 2: Distribution of age-standardized BMI by sex in school children aged 6-15 years in the Greater Accra Region of Ghana) The top graph is for girls, the bottom one for boys as indicated in the right hand lower corner in each graph.

The manuscript abstract was pasted in the online submission system and should thus not differ from the manuscript abstract.

Sincerely,
Kerstin Klipstein